vc/20-Oct-03
Week 42 starting 13 Oct 03

SM18 Operation SCHEDULE Meeting (Tuesdays)

Schedule for this week : 
	BENCH
	CURRENT

MAGNET
	STATUS 

 Today(14/10/03)
	Forward look,   & Next Magnet 

	TBA1
	3042
	ONLY FOR LEAK TEST
	Bench de-activated – preparation for D bench

	TBA2
	1020
	Final Cold Tests
	 Next Magnet 2024?

	TBC1
	3045
	Disconnection
	Next Magnet 3044 ?

	TBC2
	SSS1
	cold
	Cold tests

	TBE1
	1042
	Nearly at 1.9K

(at 1400 hrs ?)
	START COLD TESTS

	TBE2
	3039
	Disconnection
	Next 1050 ?

	TBF1
	2002 after F2 done 
	
	Prepare For 

Commissioning AFTER F2

	TBF2
	2025
	Cold
	Commissioning TESTS


Test Coordinator this week:  V Chohan;  MTM  Tech. on call:  (16-3602) : Richard

· Situation From Cryogenics:  J-P Lamboy or ACR Rep.

· Other General Issues
1. Loading up time &  AQA Analyses files(Daniel)

2. Traveller Followup – “I” shift 0830-1730

3. F Bench Commissioning 

4. Interactions in Traveller & Info  ( Q-Table, HV Tests)

5. New Test Sequence Update Web form & Display(GHH)

6. AOB

· Problems from Operation Last week (Stephane + Test Supervisor )
	Date
	Problem
	Type
	Application/soft.
	Action/advise

	Tuesday 7_M


	TBE1 : Problem with power supply.

D_2_U-L pot-aim card did not work.
	MTM
	
	Investigation by Maryline-Richard

Maybe related to cables disconnected during HV tests.

	Tuesday 7_A


	TBE2: Quench at 1.5 kA (all HF )

Triggered in ext bus bar. Symmetrical quench
	MTM
	
	Careful analysis in manual mode.

	Tuesday 7_A


	TBE2: Quench at 1.5 kA (all LF ) done in MM position.


	MTM
	
	Repeated.

	Tuesday 7_A


	TBE2: MM0 (MM position checked).

Over-range appeared in 20-21
	MTM
	
	Repeated and OK

	Wednesday 8_A


	TBE1: IAP @ warm

Card 7 in the IAP rack (2 channels ) damaged because D20 connector not plugged properly during HV test.
	MTM
	
	Follow the procedure.

	Wednesday 8_N


	MM  in TBE1 (shaft 24): Problem in the start point of the rotation
	MTM
	
	Mini-load line and tightening of shaft with TRU

Investigation on progress.

	Thursday 9_M


	MM TBE2: Angle difference in aperture 2. Integrator not properly configured.
	MTM
	
	Solved.

	Thursday 9_A


	HV test in TBF1: Problem to launch new application HV
	AB/CO
	HV
	Open vi and launch hv.vi

	Friday 10_M


	TBE1: IAP at warm not possible.

Quench heater could not be charged.
	MTM
	
	

	Friday 10_A


	LF web not starting for TBF1, 2
	
	
	

	Friday 10_A


	IAP at warm on TBE1 not launched

Mag_air_temp error.

Hf files corrupted
	MTM?
	IAP
	Done manually.

	Friday 10_A


	T coil (TBA2) cable connecting the power supply (1.5 mA) out of use.
	MTM
	
	Cable from TBE used.

	Saturday 11_M


	PLC interlock on current lead A (TBC1) .TE131 shows 547K
	MTM/ACR
	
	Correct signal simulated by Cryo.

	Saturday 11_M


	TBF2 : Connection of Shaft 

Cables missing and Tcoil cables for power supply  as well
	MTM
	
	

	Saturday 11_M
	IAP on TBA2: 3 offset could not be adjusted. 3 cards used S08_e coil which is broken
	MTM
	
	

	Saturday 11_A_M
	TBC1: Quench at 1.9 K

AQA gives time error (-48 s) for Q.heaters
	MTM
	AQA
	

	Saturday 11_M
	IAP on TBA2: 3 offset could not be adjusted. 3 cards used S08_e coil which is broken
	MTM
	
	

	Saturday 11_N
	HV at cold could not start. On TBA2 server
	AB/CO
	HV
	Run on TBA1

	Saturday 11_N
	TBA2: Main current leads not connected
	MTM
	
	Test coordinator has to foreseen this.

	Saturday 11_N
	TBC1: Tcoils errors in sector 11_12
	MTM
	
	

	Sunday 12_M
	TBA2: MM position Overrange in sector 20-21
	MTM
	
	Rotation too slow (20%)

Good at 80 %

Setting will be modified.

	Sunday 12_A
	TBA2: Quench at 1.5 kA (all HF): quench in Vbus 

Error in the YT111 heater discharges.

Repeated : same results but error in the four Qheater discharge
	MTM
	AQA
	Quench symmetrical?



	Sunday 12_A
	TBA2: Quench at 1.5 kA (all HF): repeated
	ACR
	
	Faulty trigger



	Sunday 12_N
	TBA2: Wrong current in one quench heater 


	MTM
	
	Stop the test. Investigation by Maryline.

Remove the double acquisition of LF.

Origin of the parasitic signal not clear.

	Sunday 12_N
	Training quench 1 in TBA2:  MIITS exceeded the limit
	MTM
	
	

	Sunday 12_N
	Training quench in TBA2: LF file not found automatically in file manager.
	MTM
	
	Analysis done manually.

Access with LF off and HF off done during the transfer processing?



	Monday 13_M
	Training quench 3 (TBA2): LF file not transferred.

Copy not performed correctly.

File not readable


	MTM
	
	


